Author’s response to reviews

Title: Impact of Epiphora on Vision-Related Quality of Life

Authors:

Jae-Ho Shin (pbloadsky@hanmail.net)
Yoon-Duck Kim (optho@gmail.com)
Kyung-In Woo (pbloadsky@naver.com)

Version: 3 Date: 7 August 2014

Author’s response to reviews:

Impact of epiphora on vision-related quality of life

Jae-Ho Shin1, Yoon-Duck Kim2, Kyung In Woo2 for KSOPRS3

1Department of Ophthalmology, Kyung Hee university Hospital at Gangdong, Kyung Hee University College of Medicine, Seoul, Korea
2Department of Ophthalmology, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, Korea
3Korean Society of Ophthalmic Plastic and Reconstructive Surgery

Address for correspondence: Kyung In Woo, MD. Department of Ophthalmology, Samsung Medical Center, Sungkyunkwan University School of Medicine, #81 Irwon-ro, Gangnam-gu, 135-710, Seoul, Korea. Tel: 82-2-3410-3569, Fax: 82-2-3410-3568, Email: eyeminded@skku.edu

The authors report no proprietary interest or financial support.

This was presented at Fall Scientific Symposium of American Society of Ophthalmic Plastic and Reconstructive Surgery, October 21, 2011, Orlando, FL, USA.